Author’s response to reviews

Title: Improving health information systems during an emergency: lessons and recommendations from an Ebola Treatment Centre in Sierra Leone

Authors:

Shefali Oza (shefali@alum.mit.edu)

Kevin Wing (kevin.wing@lshtm.ac.uk)

Alieu Sesay (sesayalieua@gmail.com)

Sabah Boufkhed (sabah.boufkhed@lshtm.ac.uk)

Catherine Houlihan (catherine.houlihan@doctors.org.uk)

Lahai Vandi (lkvandi@yahoo.com)

Sahr Charles Sebba (sebasahr1987@yahoo.com)

Catherine McGowan (c.mcgowan@savethechildren.org.uk)

Rachael Cummings (r.cummings@savethechildren.org.uk)

Francesco Checchi (francesco.checchi@lshtm.ac.uk)

Version: 5 Date: 13 Apr 2019

Author’s response to reviews:

Dear Dr. Duftschmid and BMC Medical Informatics and Decision Making editorial board,

Thank you for your review of our revised research article. We have responded to the editorial comments below and revised the manuscript accordingly.

Editor

1. The BMC Series does not accept panel data for publication. Please remove the panel and include the information in a different format.

We have now turned Panels 1 and 2 into tables.
2. Please place boxes over or crop the top section of each of your screenshots so that it cannot be identified which website browser is being used.

We have cropped the top section of the screenshots as suggested.

Thank you for the consideration of our manuscript and we look forward to any further feedback. Please let us know if there is any other information you need from us.

Yours sincerely,

Shefali Oza